Author's response to reviews

Title: Familial tumoral calcinosis in two Chinese patients: a case series

Authors:

Che Zhang (prof.zh@163.com)
Jiaowei Gu (giw888giw@163.com)
Xiaoli Chen (cattych163@163.com)
Kui Xiong (xiongkui@cisco.org.cn)

Version: 6 Date: 8 May 2011

Author's response to reviews: see over
Cover Letter

MS:  2094385716459480
Familial tumoral calcinosi in two Chinese patients: a case series
Che Zhang, Jiaowei Gu, Xiaoli Chen and Kui Xiong

Comments from Deputy Editor:
1. “Manuscript needs some minor English language corrections.”
Response: Some wording was modified through the manuscript, especially in case presentation section.